Author's response to reviews

Title: Did HealthKick, a randomised controlled trial primary school nutrition intervention improve dietary quality of children in low-income settings in South Africa?

Authors:

Nelia P Steyn (nelia.steyn@uct.ac.za)
Anniza de Villiers (anniza.devilliers@mrc.ac.za)
Nomonde E Gwebushe (nomonde.gwebushe@mrc.ac.za)
Catherine Draper (catherine.draper@uct.ac.za)
Jillian Hill (jillian.hill@mrc.ac.za)
Marina de Waal (mdewaal@fedicsunit.co.za)
Lucinda Dallais (luc1_za@yahoo.com)
Zulfa Abrahams (zabrahams@brhc.com)
Carl Lombard (carl.lombard@mrc.ac.za)
Estelle V Lambert (vicki.lambert@uct.ac.za)

Version: 6 Date: 11 September 2015

Author's response to reviews: see over
<table>
<thead>
<tr>
<th>Reviewers Comment</th>
<th>How it is addressed</th>
</tr>
</thead>
<tbody>
<tr>
<td>Charlotte Evans: There was one comment that still needs clarification in the text and that is the first comment by reviewer 1 on the analysis at school level. although clarification was provided in the comments in the cover letter it has not been added to the manuscript so this is still not clear enough for the reader</td>
<td>Page 8: lines 158-161: To evaluate the intervention in this cluster randomised trial an analysis was done at school level since the number of schools were small and we wished to evaluate a broader effect than at individual level only.</td>
</tr>
<tr>
<td>P B Poon: The manuscript has been much improved. I do not have further comments, except for the fact that the 3 figures provided do not appear to have Figure Legends/titles.</td>
<td>Page 20 lines 441-444 <strong>Figure titles</strong> 1. Mean dietary diversity score (DDS) over a three year intervention period 2. Mean sugar intake score (SIS) over a three year intervention period 3. Mean fat intake score (FIS) over a three year intervention period</td>
</tr>
</tbody>
</table>